Author's response to reviews

Title: STAMP alters the growth of transformed and ovarian cancer cells

Authors:

Yuanzheng He (Ajian.He@vai.org)
John A Blackford Jr (johnb@intra.niddk.nih.gov)
Elise C Kohn (kohne@mail.nih.gov)
S Stoney Simons Jr (steroids@helix.nih.gov)

Version: 8 Date: 31 March 2010

Author's response to reviews:

No response to reviewers is needed. A few minor corrections have been made to the main text of the paper.